% % Created 2024-07-10 Wed 01:09
% % Intended LaTeX compiler: pdflatex
% \documentclass[11pt]{article}
% \usepackage[utf8]{inputenc}
% \usepackage[T1]{fontenc}
% \usepackage{graphicx}
% \usepackage{longtable}
% \usepackage{wrapfig}
% \usepackage{rotating}
% \usepackage[normalem]{ulem}
% \usepackage{amsmath}
% \usepackage{amssymb}
% \usepackage{capt-of}
% \usepackage{hyperref}
% \author{Johannes Wasmer}
% \date{\today}
% \title{}
% \hypersetup{
%  pdfauthor={Johannes Wasmer},
%  pdftitle={},
%  pdfkeywords={},
%  pdfsubject={},
%  pdfcreator={Emacs 29.3 (Org mode 9.6.15)},
%  pdflang={English}}
% \begin{document}

% \tableofcontents

% \begin{center}
\begin{tabular}{llllllll}
\textbf{Model} & \textbf{Citation} & \textbf{Type} & \textbf{CSSP} & \textbf{NN} & \textbf{Defects} & \textbf{Spin} & \textbf{Symmetry}\\[0pt]
\hline
AutoML & \cite{conradBenchmarkingAutoMLRegression2022} & Composition & X &  &  & - & -\\[0pt]
\hline
Coulomb matrix & \cite{ruppFastAccurateModeling2012} & Descriptor &  &  &  & - & explicit\\[0pt]
SOAP & \cite{bartokRepresentingChemicalEnvironments2013a} & Descriptor &  &  &  & - & explicit\\[0pt]
Alchemical SOAP & \cite{lopanitsynaModelingHighentropyTransition2023} & Potential & X &  &  & - & explicit\\[0pt]
Disordered SOAP & \cite{sommer3DSCDatasetSuperconductors2023} & Descriptor &  &  & X & - & explicit\\[0pt]
GAP & \cite{bartokGaussianApproximationPotentials2010} & Potential & X &  &  & ? & explicit\\[0pt]
ACE & \cite{bochkarevEfficientParametrizationAtomic2022} & Potential &  &  &  & - & explicit\\[0pt]
\hline
MEGNetSparse & \cite{kazeevSparseRepresentationMachine2023} & Potential & X & X & X & - & explicit\\[0pt]
MACE & \cite{batatiaMACEHigherOrder2022} & Potential & X & X &  & - & explicit\\[0pt]
MACE-MP & \cite{batatiaFoundationModelAtomistic2023} & UIP & X & X &  & - & explicit\\[0pt]
\hline
Magnetic ACE & \cite{rinaldiNoncollinearMagneticAtomic2024} & Potential &  &  &  & noco & explicit\\[0pt]
SpinGNN++ & \cite{yuGeneralTimereversalEquivariant2024} & Potential & X & X &  & noco & explicit\\[0pt]
CHGNet & \cite{dengCHGNetPretrainedUniversal2023a} & UIP & X & X &  & coll & explicit\\[0pt]
SNRep & \cite{dominaSpectralNeighborRepresentation2022} & Potential &  &  &  & coll & explicit\\[0pt]
PET & \cite{pozdnyakovSmoothExactRotational2024} & Potential & ? & X &  & coll & approximate\\[0pt]
\end{tabular}
% \end{center}
% \end{document}
